\documentclass[aspectratio=169]{beamer}
\title{Presentation 1}
\author{CS460/660 - Machine Learning}
\date{27 Aug 2026}

\begin{document}
\begin{frame}{Base paper and problem setup}
\begin{itemize}
  \item Base paper and motivation
  \item What is already solved?
  \item Input and output
  \item Dataset and preprocessing
\end{itemize}
\end{frame}
\begin{frame}{Research question and tentative timeline}
\begin{itemize}
  \item What problem are you asking?
  \item Scope, assumptions, and expected contribution
  \item Tentative deliverables and dates
\end{itemize}
\end{frame}
\end{document}
